Author's response to reviews

Title: Burden of morbidity in a patient perspective - the case of sick-leave certified patients in primary care

Authors:

Lennart Carlsson (lennart.carlsson@ki.se)
Roland Morgell (roland.morgell@ki.se)
Lars-Erik Strender (lars-erik.strender@ki.se)
Britt Arrelöv (britt.arrellov@sll.se)
Gunnar H Nilsson (gunnar.nilsson@ki.se)

Version: 6 Date: 12 April 2009

Author's response to reviews:

1. A title page is included
2. Name of the Ethical Committee is written in the Methods chapter
3. Titles and legends of the two figures are embedded at the end of the main text, as is the table
4. The comments from the remaining reviewer has been taken into consideration; one sentence has been changed (ad 1: commenting Table 1) and one word has been changed (in the discussion: was said 'other', should be 'either')